Author's response to reviews

Title: Initiating decision-making conversations in palliative care: An ethnographic discourse analysis

Authors:

Emmanuelle Bélanger (emmanuellebelanger@gmail.com)
Charo Rodríguez (charo.rodriguez@mcgill.ca)
Danielle Groleau (danielle.groleau@mcgill.ca)
France Légaré (France.Legare@mfa.ulaval.ca)
Mary Ellen Macdonald (mary.macdonald@mcgill.ca)
Robert Marchand (r.marchand@sympatico.ca)

Version: 6 Date: 12 December 2014

Author's response to reviews:

December 12, 2014

The Biomed Central Editorial Team

MS: 1670763012695028 - Initiating decision-making conversations in palliative care: An ethnographic discourse analysis. Dr Emmanuelle Bélanger et al.

As requested by the editor, we stated in the methods section that all participating health care providers and patients signed informed consent forms for this study.

Best regards,

Emmanuelle Bélanger, PhD

Postdoctoral Fellow

Department of Social and Preventive Medicine, Université de Montréal & Department of Family Medicine, McGill University

7101 Parc Avenue, Montreal, Quebec, H3N 1X9

1 (514) 576-3676

emmanuellebelanger@gmail.com